Author’s response to reviews

Title: Spontaneous bladder rupture due to warfarin overdose: a case report

Authors:
Taner Sahin (drmtsahin@gmail.com)
Ufuk Oner (drufukoner@gmail.com)
Omer Baser (dr.omerbaser@gmail.com)
Ismail Kurtuncu (ismailkurtuncu@gmail.com)

Version: 1 Date: 12 Nov 2019

Author’s response to reviews:

Dear Editor

Modifications recommended by the reviewers were made. Grammar and syntax errors have been revised by the American Journal Experts. All of these changes uploaded name as "Cover Letter about Revisions".

We’d like to our case will publish on your journal. Yours sincerely.

Taner Sahin, MD
Kayseri City Hospital, Turkey
Emergency Medicine Department
drmtsahin@gmail.com